Author’s response to reviews

Title: Healthy eating and obesity prevention for preschoolers: a randomised controlled trial

Authors:

Helen Skouteris (helens@deakin.edu.au)  
Marita McCabe (marita.mccabe@deakin.edu.au)  
Boyd Swinburn (boyd.swinburn@deakin.edu.au)  
Briony Hill (briony.hill@deakin.edu.au)

Version: 5  Date: 24 April 2010

Author’s response to reviews:

The only change made to the manuscript was on page 9 lines 2&3 - we added "crèche-style, creative play activities....".